## UCSF UC San Francisco Previously Published Works

### Title

The 2020 BMT CTN Myeloma Intergroup Workshop on Immune Profiling and Minimal Residual Disease Testing in Multiple Myeloma

**Permalink** https://escholarship.org/uc/item/8t08f5rh

**Journal** Transplantation and Cellular Therapy, 27(10)

**ISSN** 2666-6375

### **Authors**

Holstein, Sarah A Bahlis, Nizar Bergsagel, P Leif <u>et al.</u>

**Publication Date** 

2021-10-01

### DOI

10.1016/j.jtct.2021.05.027

Peer reviewed



## **HHS Public Access**

Transplant Cell Ther. Author manuscript; available in PMC 2022 October 01.

Published in final edited form as:

Author manuscript

Transplant Cell Ther. 2021 October; 27(10): 807-816. doi:10.1016/j.jtct.2021.05.027.

## The 2020 BMT CTN Myeloma Intergroup Workshop on Immune Profiling and Minimal Residual Disease Testing in Multiple Myeloma

Sarah A. Holstein<sup>1</sup>, Nizar Bahlis<sup>2</sup>, P. Leif Bergsagel<sup>3</sup>, Manisha Bhutani<sup>4</sup>, Niccolo Bolli<sup>5</sup>, Carrie Brownstein<sup>6</sup>, Pierre Demolis<sup>7</sup>, David Foureau<sup>4</sup>, Francesca Gay<sup>8</sup>, Irene M. Ghobrial<sup>9</sup>, Nicole Gormley<sup>10</sup>, Jens Hillengass<sup>11</sup>, Martin Kaiser<sup>12</sup>, Marcela V. Maus<sup>13</sup>, J. Joseph Melenhorst<sup>14</sup>, Maximilian Merz<sup>11,15</sup>, Michael O'Dwyer<sup>16</sup>, Bruno Paiva<sup>17</sup>, Marcelo C. Pasquini<sup>18</sup>, Nina Shah<sup>19</sup>, Sandy W. Wong<sup>19</sup>, Saad Z. Usmani<sup>4</sup>, Philip L. McCarthy<sup>11</sup> <sup>1</sup>University of Nebraska Medical Center, Omaha, NE, USA

<sup>2</sup>University of Calgary, Arnie Charbonneau Cancer Research Institute, Calgary, Alberta, Canada

<sup>3</sup>Mayo Clinic, Scottsdale, AZ, USA

<sup>4</sup>Levine Cancer Institute, Charlotte, NC, USA

<sup>5</sup>Department of Oncology and Hemato-Oncology, University of Milan, and Hematology Unit, Fondazione IRCCS Ca' Granda Ospedale Maggiore Policlinico, Milan, Italy

<sup>6</sup>Cellectis Inc, New York, NY, USA

<sup>7</sup>European Medicines Agency, Amsterdam, Netherlands

<sup>8</sup>University of Torino, Divisione di Ematologia 1, Myeloma Unit, Azienda Ospedaliera Citta della Salute e della Scienza, Torino, Italy

<sup>9</sup>Dana-Farber Cancer Institute, Harvard Medical School, Boston, MA, USA

<sup>10</sup>U.S. Food and Drug Administration, Silver Spring, MD, USA

<sup>11</sup>Roswell Park Comprehensive Cancer Center, Buffalo, NY, USA

<sup>12</sup>Institute of Cancer Research, London, UK

- <sup>13</sup>Massachusetts General Hospital, Boston, MA, USA
- <sup>14</sup>University of Pennsylvania, Philadelphia, PA, USA
- <sup>15</sup>Department of Medicine II, University Clinic Leipzig, Germany

<sup>16</sup>National University of Ireland, Galway, Ireland

<sup>&</sup>lt;sup>\*</sup>Corresponding author: Sarah A. Holstein, MD, PhD, Division of Oncology and Hematology, Department of Internal Medicine, University of Nebraska Medical Center, Omaha, NE 68198.

**Publisher's Disclaimer:** This is a PDF file of an unedited manuscript that has been accepted for publication. As a service to our customers we are providing this early version of the manuscript. The manuscript will undergo copyediting, typesetting, and review of the resulting proof before it is published in its final form. Please note that during the production process errors may be discovered which could affect the content, and all legal disclaimers that apply to the journal pertain.

<sup>17</sup>Clinica Universidad de Navarra, Centro de Investigacion Medica Aplicada (CIMA), Instituto de Investigacion Sanitaria de Navarra (IDISNA), CIBER-ONC number CB16/12/00369, Pamplona, Spain

<sup>18</sup>Medical College of Wisconsin, Milwaukee, WI, USA

<sup>19</sup>University of California San Francisco, San Francisco, CA, USA

### Abstract

The fifth annual Blood and Marrow Transplant Clinical Trials Network (BMT CTN) Myeloma Intergroup Workshop on Immune Profiling and Minimal Residual Disease Testing in Multiple Myeloma was conducted as one of the American Society of Hematology Annual Meeting Scientific Workshops on Thursday December 3, 2020. This workshop focused on four main topics: 1) integrating MRD into clinical trial design and practice; 2) the molecular and immuno-biology of disease evolution and progression in myeloma; 3) adaptation of next generation sequencing, next generation flow cytometry and CyTOF techniques; and 4) chimeric antigen receptor T-cell and other cellular therapies for myeloma. In this report, we provide a summary of the workshop presentations and discuss future directions in the field.

### Keywords

Minimal residual disease; immune profiling; multiple myeloma; endpoint; CAR T-cell; cellular therapy

### Introduction:

Since 2016, the Blood and Marrow Transplant Clinical Trials Network (BMT CTN) Myeloma Intergroup has conducted an annual workshop focused on minimal residual disease (MRD) and immune profiling (IP) assessment in multiple myeloma (MM) which is held prior to the American Society of Hematology (ASH) annual meeting.<sup>1–4</sup> In 2019 and 2020, these workshops were formally associated with ASH as Scientific Workshops, with the 2020 workshop being conducted virtually due to the ongoing COVID-19 pandemic. The objectives of the 2020 workshop included discussing the latest scientific developments in plasma cell disorders (PCDs) with an emphasis on the immunology and molecular biology, presenting the most current clinical trial results with MRD and IP data, understanding regulatory issues regarding the use of novel endpoints in clinical trial design and drug approval, discussing novel strategies for MRD and IP testing and encouraging investigators and trainees from around the world to interact with experts in the field of PCDs.

MRD assessment of the bone marrow, most typically evaluated via either multiparametric flow cytometry (MFC) (also referred to as next generation flow (NGF)) or next generation sequencing (NGS), has been established as an important tool by which to evaluate depth of response to treatment in MM. In 2016, the International Myeloma Working Group (IMWG) published consensus guidelines for integrating MRD assessment into the response criteria.<sup>5</sup> These guidelines specified that if using MFC, then the Euro-Flow procedure<sup>6</sup> with a minimum sensitivity of  $1 \times 10^{-5}$  should be used, while if using NGS, then a

validated platform (e.g., the FDA-approved clonoSEQ assay (Adaptive Biotechnologies)) with a minimum sensitivity of  $1 \times 10^{-5}$  should be used. These guidelines moved depth of response beyond the traditional complete response (CR) or stringent complete response (sCR) categories, to include the following categories (all requiring fulfilment of CR criteria as well): flow MRD-negative, sequencing MRD-negative, imaging plus MRD-negative (flow or sequencing MRD as well as resolution of all areas of abnormal uptake on PET/CT) as well as sustained MRD-negative (MRD negativity in the marrow and by imaging confirmed a minimum of one year apart).<sup>5</sup> However, while the data are accumulating that achievement of MRD-negativity in MM is associated with superior progression free survival (PFS) or overall survival (OS) durations,<sup>7-10</sup> the field has not yet established that MRD-negativity can be used as a surrogate endpoint in clinical trials from a regulatory perspective or that MRD-response status should dictate treatment decisions. While most ongoing/planned clinical trials in MM are incorporating MRD assessment (at a minimum as an exploratory or secondary endpoint), the real-world use of MRD assessment is heterogeneous. The field is continuing to evolve, with new methodologies being developed (as discussed below) to evaluate not only bone marrow MRD but also peripheral blood MRD, as well as investigating connections between MRD status and immunophenotype<sup>11-16</sup> and MRD status in the context of novel cellular therapies.<sup>17</sup>

The 2020 workshop included 17 presentations as well as four live question and answer sessions (full agenda is shown in the Supplemental Material). In the present report we provide a comprehensive summary of the workshop, which focused on four main topics: integration of MRD into clinical trial design and practice, the molecular and immunological evolution of MM, novel detection methods, and chimeric antigen receptor (CAR) T-cell and other cellular therapies for MM.

### Session 1: Integrating MRD into Clinical Trial Design and Clinical Practice:

Sarah Holstein (University of Nebraska Medical Center) discussed the results of the 2020 pre-workshop survey. This is the fourth pre-workshop survey conducted over the past five years with the goal of better understanding the real-world practices of MRD and IP testing in MM.<sup>1, 3, 4</sup> The 2020 survey was developed by the organizers of the workshop and distributed by BMT CTN to 228 individuals from 115 centers and 6 companies or regulatory agencies. Over time there has been a trend towards increasing utilization of MRD: 70% (28/40) in 2016, 67% (16/24) in 2018, 79% (45/57) in 2019 and 89% (51/57) in 2020. Of those who responded that they are not ordering MRD in 2020, reasons provided included barriers in access to proper technology, reimbursement and lack of practice guidelines. However, five out of the six respondents who indicated that they did not order MRD testing, do utilize advanced imaging as part of the response assessment before and/or after ASCT. Of those respondents who did report measuring MRD, 75% (38/51) use MFC, 61% (31/51) use NGS, 10% use MALDI-TOF (5/51), 53% (27/51) use PET/CT and 18% (9/51) use MRI. The majority of respondents reported that the sensitivity of the MRD assay used was either  $10^{-5}$  (37%) or  $10^{-6}$  (47%). Fifty-four percent of respondents are assessing MRD in patients who have achieved either very good partial response (VGPR) or CR while 39% reported ordering it only in patients who had achieved a CR. There continues to be heterogeneity with respect to when MRD is assessed, with 41% reporting assessing it after ASCT,

25% after stem cell collection, 43% at one-year post-ASCT, 37% on an annual basis and 10% on an every six month basis post-ASCT. The 2020 survey asked questions regarding how the MRD results were incorporated into practice. Fifty-six percent of respondents reported that the results triggered a change in surveillance while only 21% reported that the results triggered a change in treatment. Of those, one respondent reported completely discontinuing therapy in MRD-negative patients, five respondents reported de-intensifying therapy in MRD-negative patients, one respondent reported switching therapy in MRD-positive patients and another noted that they discontinue treatment if there is sustained MRD negativity. Forty percent of respondents reported that the results are not used for clinical practice or are used inconsistently.

Routine use of IP has been variable across the surveys. In 2016, 35% (14/40) responded that they assessed immune reconstitution before and/or after ASCT. In 2017, 30% (7/23) responded yes, 53% (30/57) in 2019 and 41% (23/56) in 2020. Of those who reported assessing immune reconstitution in 2020, the majority assessed immunoglobulin levels (91%), while 56% reported using peripheral blood flow cytometry, 17% bone marrow flow cytometry and 35% measured vaccine titers. Six respondents reported that the immune reconstitution results triggered a change in surveillance while 7 respondents reported that the results triggered a change in treatment. The majority of respondents (84%) do not utilize Hevylite assessment.

In aggregate these survey results indicate that there is increasing real-world utilization of MRD assessment but heterogeneity remains with respect to modality, sensitivity, timing, frequency as well as whether the results are actually used to guide treatment decisions. Use of IP assessment remains variable, with most respondents not routinely performing comprehensive immune profiling. The Roswell Park group recently reported that serological response to vaccinations after ASCT correlates with PFS and OS,<sup>18</sup> and thus if future studies confirm these findings, more routine monitoring of vaccine titers may occur.

Nicole Gormley from the FDA discussed regulatory considerations surrounding novel endpoints and biomarker-driven clinical trials in myeloma. Factors that are considered when evaluating regulatory submissions include whether the MRD assessment (i.e., sample, timing, threshold) is a clinically valid biomarker for the proposed context (i.e., for the specific disease, the disease status, the type of therapy) as well as whether the MRD assay is analytically valid for the range of results that are important to the trial (https://www.fda.gov/ media/134605/download). This information must be included as part of the IND clinical trial submission, as well as information regarding the specific test method (instruments, reagents, specimen handling), how the test method was validated analytically for each specimen type, and a summary of the test's performance including accuracy, precision, specificity and sensitivity. For clinical trials using MRD assays that are not FDA approved, it is also critical that the informed consent documents indicate that the MRD assay is investigational. She also discussed the different ways that biomarkers can be used in clinical trial designs. One approach is the enrichment design, in which subjects are assessed for eligibility based on the biomarker assay. If subjects test negative then they are not included in the study, while positive subjects are randomized to different treatment strategies.<sup>19, 20</sup>

The advantages of this approach include a straightforward study design and the potential for a small sample size if the effect size is large. However, disadvantages include needing to screen a large number of patients if the prevalence of the biomarker in the population is low as well as not gaining any information about the biomarker-negative population. This approach is best used when there is very convincing data that treatment benefits are limited to the biomarker-positive subpopulation. Another approach is the stratified design in which patients are stratified based on biomarker result and then both groups (positive and negative) are randomized to treatment.<sup>19, 20</sup> The advantages of this approach include providing information about both the biomarker-positive and -negative populations while the disadvantages include that it may require more patients. This approach can be used when there are data that suggest that the treatment benefits are more likely to be effective in the biomarker-positive population but an effect cannot be ruled out in the biomarker-negative population. There are various statistical approaches for this type of stratification, including sequential testing of the various populations in different orders (e.g., testing the biomarkerpositive population and then the biomarker-negative population or first testing the overall population and then testing the biomarker-positive population) with the goal of limiting the type I error rate. It is recommended that the trial design and the statistical considerations be discussed with the FDA when considering using these types of designs. She also noted a number of additional challenges with biomarker-driven studies. The development of the assay may lag behind the development of the therapeutic such that it may be challenging to decide whether the therapeutic should be developed for the biomarker-positive population, the biomarker-negative population or the overall unselected population. The assay may need further refinement to increase its analytical performance such that early in development it might be hard to interpret due to high false positive or negative rates. If the biomarker is a continuous variable there may be uncertainty early in the development as to what the appropriate cut-off is that will result in a clinically meaningful selection of patients. An additional consideration for biomarker-driven trial designs is inclusion of interim analyses for futility, particularly in the biomarker-negative populations to limit potential exposure to ineffective therapies. Finally, she discussed the distinctions between a companion diagnostic and a complementary diagnostic. The former is a medical device that provides information that is essential for the safe and effective use of a corresponding drug/biologic product while the latter provides information to aid risk-benefit decisions for individual patients related to the use of a specific drug/biologic product and is not essential for that product's safe and effective use. Thus there are multiple regulatory considerations when using biomarkers in clinical trials, but there is the potential to help expedite drug development and ultimately improve outcomes for patients. Currently the general recommendation from the FDA is to include MRD as a key secondary endpoint with adequate statistical analysis as further development is needed before it could be used for regulatory decisions.

Pierre Démolis from the European Medicines Agency (EMA) provided the EMA perspective of integrating MRD into clinical trial design and practice. He noted that the process to develop guidelines on the use of MRD as a clinical endpoint in MM studies began in early 2018 and has thus far been a slow process. There are multiple points of agreement: MRD negativity is an important prognostic marker; outcomes for patients achieving MRD negativity are superior to simply achieving a CR; and it has the potential to be used as

an early decision basis to speed approvals when it is not optimal to await PFS or OS outcomes. However, demonstration of predictivity should come from several comparative trials, not from patients' records (not at the patient-level) and for that the differences in MRD negativity between arms as well as the differences in PFS/OS outcomes must be studied as part of the surrogacy evaluations. He noted there are a number of "tricky issues" at stake. These include timing of MRD assessment (e.g., at time of CR, before ASCT, during maintenance, etc.), whether depth or duration of response matters, whether MRD predicts outcomes independent of therapies, and whether the same level of evidence is needed for both early and late stages of MM. Questions remain as to how to determine surrogacy and whether thresholds could be used. There are also issues related to regulatory agencies evaluating a product based on MRD outcomes but then also needing to take into account the toxicity profiles of the therapy. All of these issues are under ongoing discussion within the EMA including through the Scientific Advice Working Party, the Oncology Working Party and the Committee for Medicinal Products for Human Use.

Francesca Gay (University of Torino) provided updates from the FORTE clinical trial with an emphasis on the MRD data that was presented at ASH 2020.<sup>21–23</sup> The FORTE trial is a randomized phase II study of 474 newly diagnosed transplant-eligible MM patients with three treatment arms for the first randomization: KCd (carfilzomib, cyclophosphamide, dexamethasone) induction followed by ASCT and KCd consolidation; KRd (carfilzomib, lenalidomide, dexamethasone) induction followed by ASCT and KRd consolidation; KRd induction/consolidation without ASCT (KRd12). There is a second randomization at time of maintenance to lenalidomide (R) alone vs carfilzomib + lenalidomide (KR). MRD was assessed by MFC after induction (optional), pre-maintenance in those achieving VGPR or better, and every six months during maintenance until PD. MRD was also assessed by NGS at the pre-maintenance time point in those achieving CR as well as every six months during maintenance until PD. While the rates of MRD negativity after consolidation were similar between the two KRd arms (and superior to the KCd arm),<sup>24</sup> the rate of sustained MRD negativity (defined as MRD negativity at two time points, one year apart using MFC with a sensitivity of  $10^{-5}$ ) was significantly higher in the KRd/ASCT arm (68%) compared to KRd12 (54%, p=0.02) or KCd/ASCT (45%, p<0.001).<sup>22</sup> This corresponded to superior PFS for the KRd/ASCT arm compared to the other two arms (3-yr PFS of 78% vs 66% (KRd12) vs 58% (KCd/ASCT)).<sup>21</sup> The 3-yr OS was 90% in the KRd/ASCT and KRd12 arms compared to 83% in the KCd/ASCT arm.<sup>21</sup> During the maintenance phase, a significantly higher proportion of patients (46% vs 32%, p=0.04) in the KR arm converted from MRD positive to MRD negative (MFC,  $10^{-5}$ ), which was associated with superior PFS (30-month PFS 81% (KR) vs 68% (R), p=0.026). Data were presented showing PFS outcomes by MRD status (using either MFC or NGS at  $10^{-5}$ ) and maintenance arm, with the best outcomes observed in the MRD-negative KR subgroup and the worst outcomes in the MRD-positive R subgroup. The outcomes for the MRD-negative R and the MRD-positive KR subgroups were very similar. There was strong concordance between the two techniques used to assess MRD. The prognostic impact of MRD-negativity on PFS was observed across multiple subgroups, including age, ISS, standard and high risk FISH, R-ISS and LDH. Benefit of achieving MRD negativity was also observed in patients with circulating plasma cells at baseline. Overall these data highlight the importance of not only assessing MRD

negativity at discrete time points, but assessing sustained MRD negativity in the context of correlation with PFS outcomes.

# Session 2: Molecular and Immunobiology of Disease Evolution and Progression in Multiple Myeloma:

This session focused on recent translational studies interrogating the molecular and immunological landscape of PCDs. Niccolò Bolli (University of Milan) discussed the molecular and genomic evolution of MM. He noted that currently the field's view of PCDs is that we do not treat patients with MGUS, consider treating patients with smoldering MM (SMM) and treat patients with MM. However, should we be treating a disease that is biologically aggressive independent of the disease burden and before evident clinical sequelae? Are we able to distinguish indolent vs aggressive disease and if so, should we be treating based on that classification and not the historical diagnosis? In work presented at ASH 2020, whole genome sequencing was performed on CD138-positive bone marrow mononuclear cells from MGUS and SMM patients and compared with samples from patients with MM.<sup>25</sup> These studies revealed that patients with non-progressive MGUS/SMM had a distinct genomic profile which lacked most of the key genomic MM hallmarks compared to the progressive patients whose disease already had many of the genomic drivers associated with MM.<sup>25</sup> Previous work investigating modes of evolution of SMM showed two routes to MM: one characterized by continuous evolution during which subclones are gained and lost, generally starting indolently and taking years to become aggressive vs the static progression mode which is genomically aggressive from the start and only needs to accumulate to a large enough burden to meet criteria for MM.<sup>26</sup> It is postulated that the latter represents a category of SMM which should be treated as active MM from time of diagnosis. Additional groups have reported that incorporation of genetic features can improve prognostication of SMM.<sup>27, 28</sup> Thus it is posited that future classification of PCDs will no longer include SMM and instead include indolent (asymptomatic) gammopathy vs aggressive (symptomatic) MM which would be determined using genomic data or other criteria.29

Irene Ghobrial (Dana-Farber Cancer Institute) discussed the immune microenvironment in MM. She noted that the microenvironment is critical for the growth of tumor cells, contributing to the progression/evolution from MGUS to MM. Single cell RNA sequencing (scRNA seq) analysis of healthy, MGUS, SMM and MM plasma cells and immune cells revealed that there are detectable changes in the composition of the immune cells even at the MGUS stage.<sup>30</sup> During the evolution to MM there is memory CD8+ T-cell depletion as well as transcriptional changes involved with the IFN-α response leading to immune escape.<sup>30</sup> Studies using 10x Genomics analysis of the BM microenvironment are also revealing compositional changes in the immune cells across the PCD spectrum. These immune alterations are heterogeneous but differences are observed between healthy controls and MGUS samples. Differences in NK cells across the disease spectrum have also been described.<sup>31, 32</sup> Further T-cell characterization shows loss of memory CD4 cytotoxic T-cells during progression along with increasing Tregs. These changes might have therapeutic implications if they could be reversed. Previous work in a mouse model showed that if Tregs

are depleted this can prevent MM progression.<sup>33</sup> Their group has also found an enhanced interferon response in multiple cell subtypes, including NK cells, CD14+ monocytes and CD16+ monocytes. Single cell RNA (scRNA) studies in the CD14+ monocytes revealed increases in MHC II mRNA levels. However, cytometry by time of flight (CyTOF) studies showed lower levels of these molecules as a consequence of regulation of surface trafficking, thus highlighting the importance of validating discoveries based on RNA data at the protein level. Ultimately the hope is that understanding the compositional changes in the immune microenvironment in patients with PCDs will lead to therapeutic interventions that can prevent progression to active MM.

Maximilian Merz (Roswell Park Comprehensive Cancer Center) discussed the use of single cell sequencing to evaluate intraclonal myeloma heterogeneity. He noted that scRNA seq analysis is one of the fastest developing techniques but while there are now multiple commercial kits available, the bottleneck can be the downstream analysis. Several groups have shown the feasibility of scRNA analysis in MM.<sup>30, 34</sup> His group is conducting a study in which paired biopsies from bone marrow and osteolytic lesions are analyzed to determine whether there are differences in the myeloma cells, the surrounding microenvironment and whether there are longitudinal changes during therapy.<sup>35</sup> The plasma cell compartment is subjected to scRNA-seq and whole exome sequencing (WES) while the non-plasma cell compartment is subjected to NGF and bulk T-cell receptor sequencing. The scRNA-seq occurs same day to eliminate any transcriptional changes that happen during the freezing process. More than 70,000 cells are isolated from each location and thus far 10 patients have been enrolled (7 newly diagnosed MM, 3 relapsed/refractory MM). His group has compared the gene expression data from the scRNAseq and variant allele frequencies and found significant correlations between the plasma cells from the bone marrow and from those isolated from the lytic lesions for all patients. Of note, in one patient with relapsed/ refractory MM and extramedullary disease, significant differences based on location were found. These studies are enabling dissection of the spatial heterogeneity of the disease at single cell resolution and allowing identification of longitudinal transcriptional changes in plasma cells in response to therapy.

Leif Bergsagel (Mayo Clinic, Scottsdale) discussed the role of the gut microbiota in shaping the evolution of MM. Clues that the gut microbiota might influence MM disease biology were obtained from studies utilizing the Vk\*MYC transgenic mice that spontaneously develop monoclonal gammopathy that eventually progresses to MM. It was noted that the rate of progression to MM in this model was different in mice housed in a pathogen-free vivarium in Italy compared to mice bred in Arizona.<sup>36</sup> While the incidence of monoclonal gammopathy remained constant, the progression to MM differed. In studies using the transplantable MM cell line Vk12598 it was found that if recipient mice were pretreated with antibiotics, engraftment of the myeloma cells was significantly delayed and survival significantly prolonged.<sup>36</sup> Analysis of the fecal microbiota from the mice that more rapidly progressed led to identification of Prevotella, which was absent in the mice from Italy. Subsequently, studies with the Vk12598 transplantation model in which mice were administered Prevotella via oral gavage, showed that these mice had faster engraftment and shorter survival than the vehicle control animals.<sup>36</sup> This work led to the hypothesis that the presence of bacteria such as segmented filamentous bacteria or Prevotella which

Page 9

are associated with high levels of Th17<sup>37</sup>, contribute to progression.<sup>36</sup> In this model, the Th17 cells which are stimulated by the bacteria in the gut then home to the bone marrow where they can stimulate MM cells through IL-17 as well as stimulate eosinophils which secrete IL-6 and promote MM progression.<sup>36</sup> It is noted that SMM patients that progressed to MM more quickly (in this case, used a 3-yr cut-off) had higher levels of IL-17 in the bone marrow than those who progressed more slowly. The gut microbiome has also been implicated in the post-ASCT setting in MM. Work by Pianko et al., identified butyrateproducing microbiota as being associated with achievement of MRD-negativity at the threemonth post-ASCT time-point.<sup>38</sup> These bacteria have been associated with decreased Th17 cell production. Potentially related to the gut microbiome story is a study that was conducted in asymptomatic MM patients in which they received the IL1-Ra agent Anakinra for 6 months.<sup>39</sup> This study demonstrated that a decrease in CRP was associated with prolonged progression to MM.<sup>39</sup> Thus in aggregate these studies suggest that the gut microbiome and inflammatory microenvironment are linked to MM progression. It is possible that modulation of the microbiome (e.g., with probiotics, antibiotics, fecal transplant, diet) with or without anti-inflammatory agents (e.g., anti-IL1β, IL1-RA, anti-IL6, anti-IL17) could delay or prevent progression to MM.

### Session 3: Adaptation of Next Generation Sequencing, Next Generation Flow Cytometry, and CyTOF: diverse ways of detection.

This session focused on the use of novel techniques to assess MRD in MM. Bruno Paiva (Clinica Universidad de Navarra) discussed issues related to the use of flow cytometry in measuring MRD and IP in MM. As has been reported now in multiple studies, there is a difference in PFS in patients achieving MRD-negativity vs MRD-positivity. In a recent analysis of the PETHEMA/GEM2012MENOS65 trial, achievement of MRD-negativity (as assessed by NGF with median limit of detection  $2.9 \times 10^{-6}$ ), was associated with significant reduction in risk of progression or death and overcame poor prognostic features at diagnosis, including high-risk cytogenetics.<sup>40</sup> He advocated for the use of NGF in assessing MRD status as this technique allows for evaluation of sample quality (e.g., assessment of hemodilution) as well as simultaneous immune profiling of the tumor microenvironment. As an example, in a subset of patients from the PETHEMA/GEM2012MENOS65 study, immune monitoring studies were performed using 17-color NGF as well as combined scRNA/TCR sequencing. These studies revealed that patients with higher CD27-/+ T-cell ratios have prolonged PFS irrespective of MRD status and that this parameter was a surrogate for how many clonotypic T-cells were present in the bone marrow.<sup>41</sup> A future is envisioned in which this type of analysis is routinely performed, allowing prediction of outcomes of patients exposed to specific therapies based on the MRD level, immune composition, patient demographics, staging, cytogenetics and prior therapies.

David Foureau (Levine Cancer Institute) discussed the use of mass spectrometry (MS) as a potential measure of MRD, reporting on the results of a study in which 10-marker NGF MRD testing (bone marrow aspirate) was compared with MS-based MRD testing (peripheral blood) in order to compare sensitivity and concordance rates. The MS protocol involves an immuno purification step (to assess IgG, IgA, IgM, total light chains and

free light chains) followed by matrix-associated laser desorption/ionization (MALDI)-MS or liquid chromatograph (LC)-MS to detect the monoclonal protein. A diagnostic sample is required to define the original monoclonal protein for the MS method. Twenty-eight matched bone marrow aspirate and serum (collected within 2 weeks of each other) specimens were analyzed. In six patients with kappa free light chain disease (as defined by serum immunofixation electrophoresis (SIFE)), the MALDI-MS method could also identify a heavy chain. In addition, MALDI-MS could detect therapeutic antibodies (e.g., daratumumab) as well as glycosylated monoclonal proteins. Concordance rates between the two methods were compared at three levels of sensitivity (for NGF;  $10^{-4}$ ,  $10^{-5}$ ,  $10^{-6}$ ) and were found to be in the 60–70% range. This is comparable to previously reported rates from other groups.<sup>42, 43</sup> However, in this cohort none of the samples were negative by LC-MS, thus it is unclear whether this technique has higher sensitivity or lower specificity. Several examples were discussed where a patient was negative for MRD by NGF ( $10^{-6}$ ) but positive by MALDI-MS or LC-MS. However, another example was discussed where there was discordance between NGF ( $10^{-6}$ ) and MALDI-MS at a one-year post-ASCT time point because the glycosylated monoclonal protein identified at diagnosis became unglycosylated at the later time point (thus MALDI-MS was reported as negative) while NGF could still detect residual clonal cells. Thus MS-MRD may be more sensitive than NGF in some patients (perhaps because it is not dependent on marrow involvement), but NGF can detect residual disease independent of monoclonal protein glycosylation.

Martin Kaiser (Institute of Cancer Research) discussed MRD assessment in the context of myeloma genetics and clinical trial design. He noted that many planned/ongoing studies are assessing MRD at a fixed time point and then making treatment decisions based on the results (e.g., escalation, continuation, de-escalation). However, the question is raised as to whether this type of approach adequately addresses different disease biologies. He noted a study evaluating high-risk MM using gene expression profiling and cytogenetic information in which an ultra-high risk subgroup was identified which has a higher proliferative capacity than the other high risk groups and had early relapse.<sup>44</sup> This is of note because if studies are designed evaluating high-risk MM but there is a significant number of patients progressing during induction before the MRD-driven treatment decision point is reached, then only a fraction of patients would benefit from treatment escalation if they are MRD-positive. It is also important to note that there have been examples of false negative MRD results in patients with ultra-risk disease where whole body diffusion weighted MRI can detect residual disease. In this context, the UK has designed clinical trials which combine both MRD-adapted approaches and risk-adapted approaches. For example, in the UK-MRA Myeloma XV study, patients with ultra-high risk disease do not undergo MRD-adapted treatment while those without ultra-high risk disease have treatment which is dictated by MRD status post-ASCT. Thus while MRD-negativity is desirable for all risk groups, it is important that MRD-adapted therapy consider disease heterogeneity.

Nizar Bahlis (University of Calgary) discussed single cell immune profiling in myeloma. Previously reported studies using single cell profiling have provided detailed views and insight into immune cell populations, clonal heterogeneity, and tumor ecosystem temporalspatial dynamics.<sup>45–48</sup> Epigenetic analysis at the single cell level in conjunction with transcriptome analysis using projectory inference can better define the origin and trajectory

of these key immune cells. In an analysis of MM patients treated with daratumumab +/pomalidomide, it was determined that daratumumab- and immunomodulatory drug (IMiD)sensitive patients have different immunomes than resistant patients, with bone marrow enrichment of CD8+C27+TEM cells noted in the sensitive patients.<sup>49</sup> In contrast, resistant patients had an enrichment in exhausted T cells with high expression of the checkpoint inhibitors LAG3 and TIGIT, granzyme B and, at the transcriptional level, high expression of TBX21 and lack of TCF7. In addition to this T cell deregulation, the dendritic cells were largely immature, lacking class II MHC molecules and the ratio of dendritic cells to Tregs was significantly higher in sensitive patients vs resistant. These data suggest that the immature dendritic cells in resistant patients results in ineffective priming of CD4 helper T cells which leads to a dysfunctional cytotoxic T cell response, and that the chronic stimulation results in exhausted T cells. They have also characterized the mononuclear cells in patients receiving B-cell maturation antigen (BCMA) CAR T-cell therapy as well as BCMA bispecific T-cell directing therapy. In an patient that failed to respond to a bispecific agent, cellular indexing of transcriptomes and epitopes (CITE)-seq profiling revealed that the CD8 T-cells were mostly CD45RA-positive, lacked selectin and CD28 and had high expression of senescence and exhaustion markers.<sup>50</sup> A model arises where sensitive patients have an immunity phenotype (reminiscent of MGUS) whereas resistant patients have a tolerance phenotype (reminiscent of symptomatic MM).<sup>30, 51</sup> It is hypothesized that the expansion of terminally exhausted T cells with upregulation of checkpoint inhibitors and enrichment of immature tolerogenic dendritic cells mediate resistance to immune-based interventions. It is hoped that further single cell studies at the level of the transcriptome, proteo-metabolome and epigenome will eventually allow for therapeutic interventions that can reverse tolerance.

### Session 4: CAR-T and other Cellular Therapy for Multiple Myeloma:

This session focused on novel cellular therapies for myeloma, including NK cells, allogeneic CAR T-cells and autologous CAR T-cells. Michael O'Dwyer (National University of Ireland) discussed the potential for NK cells to serve as an allogeneic cell source. NK cells are dysfunctional in MM<sup>52</sup>, and while certain therapies (e.g., IMiDs) can improve NK cell function they may not completely reverse it. In addition, treatment with anti-CD38 monoclonal antibodies (mAbs) can lead to prolonged depletion of NK cells. Therefore allogeneic NK cell therapy may be an attractive alternative which allows for off-the-shelf therapy. The potential sources for the NK cells include umbilical cord blood, healthy donors, NK cell-derived cell lines or stem cells. Following enrichment and/or differentiation, along with T-cell depletion, NK cells undergo extensive expansion prior to infusion or cryopreservation. These expanded NK cells can be genetically modified or alternatively, progenitors, such as induced pluripotent stem cells, can be genetically engineered prior to subsequent differentiation into mature NK cells. Promising preliminary data have emerged from clinical trials, including the trial conducted by Shah et al., supporting the safety and potential efficacy of allogeneic NK cells in MM.<sup>53</sup> In addition, it is noted that allogeneic NK cells can be combined with mAbs and NK-engaging molecules to further enhance their targeting and cytotoxicity. There are many ways that NK cells can be engineered to optimize their function, cytotoxicity, persistence and homing.<sup>54</sup> Using gene editing

(e.g., with CRISPR/Cas9) it is now possible to routinely knock out inhibitory receptors in primary NK cells with high efficiency improving their function.<sup>55</sup> Another example includes knocking out CD38 to avoid fratricide when targeting CD38.56 This modification has been shown to not only enhance antibody-dependent cellular cytotoxicity (ADCC) in the presence of anti-CD38 antibodies but also to enhance the metabolic profile of NK cells.<sup>56</sup> NK cells can also be engineered to express CARs along with a potent form of TRAIL to maximize apoptosis via the death receptor pathway and minimize antigen escape (ONK Therapeutics). Overall, engineered NK cells may have several advantages over the autologous CAR T-cell approach.54 The engineered NK cells are allogeneic, frozen, off-the shelf, dependent on innate killing and independent of antigen. While historically NK cells have been more difficult to engineer, recent advances have overcome many of the limitations. Thus far, NK cell therapy has not been associated with CRS or neurotoxicity and therefore have the potential for outpatient administration. It was noted that encouraging responses were reported in a CD19 B cell malignancy trial involving CAR NK cell therapy.<sup>57</sup> The persistence of NK cells may be a limitation as traditionally, allogeneic NK cells have not been detected beyond approximately two weeks. However, with engineering, persistence can be much longer and in the work by Liu et al, they were able to detect the presence of the construct via PCR in the peripheral blood after a year.<sup>57</sup> Whether the duration of NK cell persistence needs to be the same as CAR T-cell persistence remains to be determined.<sup>57</sup> While engineered NK cell therapy is first being tested in the relapsed/refractory MM setting, it does have the potential to be moved to the frontline setting, possibly for patients with high risk cytogenetics or persistent MRD-positivity. Indeed, given their potential to efficiently target clonogenic MM cells,<sup>58</sup> NK cells may be particularly effective in eradication of MRD.

Carrie Brownstein (Cellectis Inc) discussed allogeneic CAR T-cell therapy. She noted that allogeneic cells offer several potential advantages over autologous cells—they can be readily available, may be produced more reliably and on a greater scale. The use of allogeneic cells avoids the manufacturing delays and/or failures sometimes associated with an autologous product. Other potential advantages of using allogeneic cells include the use of healthy donor T cells which may confer better clinical activity, enhanced standardization as well as the opportunity for re-dosing. Finally, there may be an advantage from a cost-effectiveness perspective if multiple doses can be produced from one donor. Cellectis is focusing on the production of allogeneic products using the TALEN<sup>®</sup> gene-editing tool. The manufacturing process starts with healthy donor leukopaks that are transduced with a lentivector and gene edited by TALEN. The CAR T-cells are then amplified, purified and frozen. She discussed the UCARTCS1A product which recognizes MM cells via the anti-CS1 CAR. Potential advantages of the CS1 CAR product include lymphocyte CS1 expression, which could result in extended lymphodepletion and enhanced CAR T-cell persistence. The TALEN gene editing tool allows for an engineered product that lacks CS1 (to avoid fratricide) as well as the T-cell receptor (to avoid graft-vs-host disease) to improve the yield of CD8+ cells, creating a less differentiated T-cell phenotype and resulting in higher *in vitro* anti-tumor activity. This product also expresses a CD20 mimotope (RQR8) that serves as a safety switch responsive to rituximab treatment. What remains to be determined is how long the allogeneic CAR T-cells need to persist in order to confer a durable remission. As CS1 is expressed on lymphocytes in addition to MM cells, finding the optimal window

of UCARTCS1A expression and persistence that will induce MRD negativity without longterm lymphopenia is critical. The first-in-human study (MELANI-01) is evaluating the safety and tolerability of the UCARTCS1A product in triple-class refractory MM patients. The study was placed on clinical hold in July 2020 after a subject experienced a fatal cardiac arrest. In November 2020 the FDA lifted the clinical hold and the study is currently ongoing.

Marcela Maus (Massachusetts General Hospital) discussed the approach her research group is taking in developing novel autologous CAR T-cell therapies. She noted that current anti-BCMA CAR T-cell therapies are not curative and that this may be because they contain non-human sequences that can be rejected or there is loss of BCMA expression on the target cell. APRIL (a proliferation-inducing ligand) is produced by myeloid cells within the bone marrow microenvironment and promotes survival of MM cells via binding to BCMA and TACI (transmembrane activator and calcium modulator and cyclophilin ligand interactor). Soluble APRIL forms a trimer which can then interact with BCMA and TACI. Her group is designing CARs that use a fully human trimeric APRIL that can bind to two different antigens on the plasma cell simultaneously. They have evaluated both a monomeric APRIL fused to the CAR backbone as well as a trimeric APRIL CAR (TriPRIL).<sup>59</sup> The truncated monomeric APRIL had very weak binding to BCMA and negligible binding to TACI. However, the TriPRIL CAR, which is composed of the APRIL domains connected via linker domains, has enhanced binding to both BCMA and TACI.<sup>59</sup> Evaluation of TriPRIL in *in* vitro studies revealed enhanced functionality (measured via single cell cytokine analysis) compared with the monomeric APRIL CAR or a BCMA CAR. TriPRIL was demonstrated to lyse primary MM cells.<sup>59</sup> Finally, in xenograft studies they were able to demonstrate activity of the TriPRIL CAR T cells in vivo in both BCMA-positive and BCMA-negative MM models.<sup>59</sup> Efforts are now underway to move this novel cellular therapy to the clinic. A GMP vector has been made and the investigators are in the process of scaling up manufacturing using the CliniMACS Prodigy<sup>®</sup> system. A phase I trial is expected to begin enrolling in 2021. Notably, the study will include patients who have previously received BCMA-directed therapy.

J. Joseph Melenhorst (University of Pennsylvania) discussed IP studies being conducted in both responders and non-responders to CAR T-cell therapy in a variety of hematological malignancies. He noted that while in acute lymphoblastic leukemia (ALL) the responses to CD19 CAR T-cell therapy are quite high,<sup>60</sup> the responses (using the same CAR) are much lower in chronic lymphocytic leukemia (CLL).<sup>61</sup> In a study evaluating CTL019-treated CLL patients, it was found that responders to this therapy had higher CD27 and lower CD45RO levels in the peripheral blood T-cells that were used in manufacturing.<sup>62</sup> This CD27+CD45RO- biomarker was found to have 89% specificity and 80% sensitivity in the discovery cohort and 100% specificity/sensitivity in the validation cohort for identifying CR patients.<sup>62</sup> More recently his group has been performing fate mapping studies of CD19 CAR T-cells. Two patients with CLL treated with CTL019 in July 2010 have been found to have persistent CAR T-cells, 7-10 years after infusion, and remain in remission. Using a 40-marker CyTOF panel, the cells can be used to interrogate T-cell differentiation, activation, exhaustion and the anti-CAR19 idiotype mAb. This analysis showed multiple different clusters separated by CD4 and CD8 expression which varied in dominance over time. Over time, the CD4+ CAR T-cells gradually dominated the CAR T-cell repertoire

in both patients, suggesting a role for these cells in sustained remissions. The pattern and tempo of the clonal evolution differed between the two patients, but in both patients at later time points there was an oligoclonal repertoire suggesting that the remission is being sustained by a few persisting clones. Studies are ongoing to evaluate the hypothesis that the integration site potentially disrupts endogenous genes and thereby affects the potency of the CAR T-cells. Studies are also underway to identify clusters enriched in responding and non-responding patients to determine whether the CLL biomarker profile can be validated in other malignancies such as MM. Previously published work from this group revealed the impact that prior therapies have on the percent of T-cells with the CD8+CD27+CD45RO– phenotype, with higher levels post-induction but then lower levels in the relapsed/refractory setting.<sup>63</sup> Finally, it was noted that the species of CAR used may have an impact on the re-expansion of the cells as four patients who had previously been treated with a murine BCMA CAR T-cell and relapsed subsequently responded to a fully human BCMA CAR.<sup>64</sup>

Sandy Wong (University of California San Francisco) discussed the clinical laboratory predictors of durable response to CAR T-cell therapy. She noted that the traditional IMWG criteria rely on clearance of monoclonal protein, but because of the half-life of antibodies, this can lag behind cell killing. CAR T-cell treatment can induce dramatic early responses. Therefore the question is raised whether there are other markers which could be used to evaluate an early response that is also associated with PFS. She presented an analysis of 54 patients treated at her institution on five different industry-sponsored BCMA CAR T-cell trials. An analysis of PFS by IMWG response in this cohort showed no statistically significant difference between those patients achieving partial response/VGPR and those achieving a CR. However, normalization of the involved free light chain (FLC) at either day 15 or day 30 was associated with improved PFS. Achievement of MRD negativity  $(10^{-6})$ by NGS at month 1 or month 3 was associated with improved PFS as was achievement of MRD negativity (10<sup>-5</sup>) or PET/CT negativity. In this cohort, the occurrence of CRS and neurotoxicity was not associated with PFS. Larger prospective studies are needed to determine whether normalization of FLC at day 15 or MRD-negativity at 1 month could be validated as predictors of durable response to CAR T-cell therapy in MM.

### Future directions:

The role for MRD assessment in determining clinical trial design, regulatory evaluation and real-world treatment decisions is continuing to evolve. While it is most likely that initial establishment of MRD as a surrogate endpoint will be based on MFC/NGF assessment of bone marrow aspirate samples, much uncertainty remains as to the optimal timing of MRD assessment as well as the optimal definition of sustained MRD. The FORTE study is a prime example of the importance of not forming conclusions regarding relative efficacy of different treatments based on one MRD assessment point alone. As noted above, while the MRD negativity results for the two KRd arms were similar post-consolidation, it is the rate of sustained MRD negativity which is now correlating with PFS outcomes, demonstrating superiority of KRd/ASCT over KRd12. Thus the evaluation of multiple time points is likely critical. The potential to move beyond marrow-based assessments is significant, as this would allow more frequent assessments in a non-invasive manner and can capture evidence of disease that is outside of the marrow. The role of MS-based monitoring of residual

monoclonal proteins is continuing to evolve but appears poised to become an important complement to marrow-based MRD assays.

As noted by Dr. Gormley, there are a number of challenges as well as potential benefits of using biomarker-driven trial designs. One key point made by Dr. Kaiser is that disease biology also needs to be accounted for, and that studies utilizing MRD status to determine treatment path may exclude patients with aggressive MM that progress prior to ever reaching the MRD-guided time point. Thus design of clinical trials incorporating MRD assessment in MM require substantial input from disease specialists, statisticians and regulatory agencies. The results from the 2020 pre-workshop survey again revealed heterogeneity in real-world use of MRD assessment. Until there are data demonstrating that the results of MRD testing can actually be used to change treatment practices (with resulting improvement in long-term outcomes), it is likely that practitioners will continue to make either anecdotally-driven treatment decisions (e.g., treatment was stopped in a patient with MRD-negativity and they are doing fine, therefore that should form the basis of routine practice) or they will collect the MRD data for use in future retrospective studies. The latter approach may prove helpful, as there remains an absence of high quality data supporting the use of MRD-adapted treatment approaches.

The exploration of the MM immune microenvironment is becoming more sophisticated with multiple groups now evaluating the immunome at the cellular level. As discussed in this workshop, different signatures have been identified in different disease contexts and it is likely that while there may be some unifying themes (e.g., association with T-cell exhaustion/senescence phenotypes and PCD progression and/or resistance), some of these phenotypes will be specific to discrete disease states and/or therapies. A more consistent incorporation of comprehensive IP studies in all prospective MM therapeutic intervention trials would provide a wealth of information. This would require not only consensus within the field regarding optimal IP techniques (e.g., MFC vs CyTOF vs scRNAseq vs other) but also commitment from sponsors to pay for these correlative studies. In the long run this could prove cost-effective if phenotypes are identified that may predict for sensitivity (or resistance) to high-cost therapies such as cellular or T-cell engaging therapies as well as expensive multi-agent combination therapies. Another layer of complexity is the role of the gut microbiome in modulating PCD progression, responsiveness to therapies and toxicities. While most recent studies have focused on the role of the gut microbiome in the hematopoietic stem cell transplant setting, 65-67 there is emerging work exploring the association between the microbiome and CAR T-cell therapy.<sup>68, 69</sup> Thus with the development of not only autologous CAR T-cell but also allogeneic CAR T-cell and NK cell therapy for MM, there is significant opportunity to perform comprehensive IP and microbiome studies to better understand treatment efficacy and toxicity. The long-term effects of autologous CAR T-cell therapy are still being discovered<sup>70</sup> and it remains to be determined whether NK cell<sup>71</sup> or allogeneic CAR T-cell therapies will have distinct short-term and/or long-term toxicities. Finally, the use of simpler, more readily available assessments of immune reconstitution such as seroconversion following vaccine administration post-ASCT (particularly in the COVID-19 pandemic era) should also be considered.18

The continued progress being made in dissecting the genetic and immunological underpinnings of PCDs should ultimately lead to therapeutic interventions that may delay/ prevent progression of PCDs as well as new therapeutic strategies for the management of MM. In addition, identification and incorporation of novel biomarkers into clinical trial designs should enable precision medicine-based approaches for MM. Progress is continuing to be made in establishing MRD as a surrogate endpoint, with the hope that this could accelerate drug approval timelines. However, surrogacy has not yet been established and many questions remain regarding the requisite duration of response, depth of response and timing of response. Ultimately the development of evidence-based guidelines for the use of MRD and IP in real-world practice will depend on continued collaborative efforts amongst researchers, industry sponsors and regulatory agencies.

### Supplementary Material

Refer to Web version on PubMed Central for supplementary material.

### Acknowledgements:

Support for the Blood and Marrow Transplant Clinical Trials Network was provided by grant #U10HL069294 and #U24HL138660 from the National Heart, Lung, and Blood Institute (NHLBI) and the National Cancer Institute (NCI). The content is solely the responsibility of the authors and do not reflect the views or the official policy or position of the National Institutes of Health. This article reflects the views of the authors should not be construed to represent the FDA's views or policies.

#### Conflict of Interest:

SAH: has served on advisory boards for Genentech, GlaxoSmithKline, Oncopeptides, Sanofi, Takeda; has served as a consultant for Celgene, Sorrento; has received research funding from Oncopeptides.

NB: has received honoraria from Janssen, Celgene, Amgen, Takeda, AbbVie, GlaxoSmithKline, Sanofi, Genentech and Karyopharm; has served as a consultant/advisor for Janssen, Celgene, Amgen, Takeda, AbbVie, GlaxoSmithKline, Karyopharm, and Genentech

PLB: has served as a consultant for Bristol-Myers Squibb, Janssen, Amgen and Novartis

MB: has served on speaker bureaus for Amgen, Bristol-Myers Squibb, Takeda; has served as a consultant for Sanofi Genzyme; has received research funding from Celgene, Bristol-Myers Squibb, Cerecor, Cellularity, Janssen, MedImmune, Takeda, Prothena

NB: has received honoraria from Janssen, Celgene, Bristol-Myers Squibb, Amgen, Takeda; has served on advisory boards for Janssen, Celgene

CB: is a full-time employee of Cellectis, Inc.

PD: has nothing to disclose

FG: has received honoraria from Janssen, Celgene, Bristol-Myers Squibb, Amgen, Takeda; has served on advisory boards for Janssen, Celgene, Bristol-Myers Squibb, Amgen, Takeda, Adaptive Biotechnologies, Roche, Abbvie and Oncopeptides.

IMG: has served as a consultant or advisor for GlaxoSmithKline, Sanofi, Janssen, Takeda, Karyopharm, AbbVie, GNS, Cellectar, Adaptive, Bristol-Myers Squibb, Oncopeptides.

NG: has nothing to disclose

DF: has received research funding from Celgene and TeneoBio

JH: has received honoraria from Janssen; has served on advisory boards for Adaptive Biotechnologies, Amgen, AXXESS Network, Bristol-Myers Squibb, Celgene, GlaxoSmithKline, Janssen, Oncopeptides, Oncotracker, Sanofi and Skyline.

MK: has served as a consultant for Amgen, Bristol-Myers Squibb/Celgene, GlaxoSmithKline, Janssen, Karyopharm, Seattle Genetics, Takeda; has received honoraria from Bristol-Myers Squibb/Celgene, Janssen, Takeda; has received travel/educational support from Bristol-Myers Squibb/Celgene, Janssen, Takeda; has received research support to institution from Bristol-Myers Squibb/Celgene

MVM: is an inventor on patents related to adoptive cell therapies held by Massachusetts General Hospital and University of Pennsylvnia (some licensed to Novartis). She holds equity in Ichnos, TCR2 and Century Therapeutics, and has served as a consultant for Adaptimmune, Agenus, Allogene, Arcellx, Astellas, AstraZeneca, Atara, Bayer, BMS, Cabaletta Bio (SAB), Cellectis (SAB), CRISPR therapeutics, In8bio (SAB), Innovakine, Intellia, GlaxoSmithKline, Kite Pharma, Micromedicine, Novartis, TCR2 (SAB), Tmunity, Torque, and WindMIL (SAB). She is on the Board of Directors for Ichnos Science.

JJM: has received research funding from IASO Biotherapeutics and Kite Pharma, a Gilead company. He has served as speaker for Novartis and Johnson & Johnson and consults for Simcere of America, Shanghai Unicar Therapy, Janssen Research & Development, LLC, Poseida, Allogene, and IASO Biotherapeutics. He is on the Medical and Scientific Advisory Board of IASO Biotherapeutics. He holds patents related to CAR T cell manufacturing and biomarkers.

MM: has nothing to disclose

MO: has served as a consultant and advisor for Janssen, Abbvie, Amgen, Celgene, Adaptive Biotech, Glycomimetics and Carrick Therapeutics; has received research support from Janssen, BMS, Abbvie and ONK Therapeutics; has equity ownership in Carrick Therapeutics and ONK Therapeutics; is employed by ONK Therapeutics.

BP: has nothing to disclose

MCP: has served as a consultant for Bristol-Myers Squibb and Amgen; has received research support from Kite, Novartis and Bristol-Myers Squibb

NS: has received research funding from Celgene/Bristol-Myers Squibb, Janssen, Bluebird Bio, Sutro biopharma, Teneobio, Poseida, Nektar; has served as an advisor for GlaxoSmithKline, Amgen, Indapta Therapeutics, Sanofi, CareDx, Kite, Karyopharm, Oncopeptides and CSL Behring.

SWW: has served as a consultant for Amgen; has served on advisory board for Sanofi; has received research support from Janssen, GlaxoSmithKline, Genentech and Fortis

SZU: has received research funding from Amgen, Array Biopharma, Bristol-Myers Squibb, Celgene, GlaxoSmithKline, Janssen, Merck, Pharmacyclics, Sanofi, Seattle Genetics, SkylineDX, Takeda; has received consulting fees from Amgen, Bristol-Myers Squibb, Celgene, GlaxoSmithKline, Janssen, Merck, Sanofi, SkylineDx, Takeda and has received speaking fees from Amgen, Celgene, Janssen and Takeda.

PLM: has served as a consultant for Bird Biotech, Bristol-Myers Squibb, Celgene, Fate Therapeutics, Janssen, Juno, Karyopharm, Magenta Therapeutics, Sanofi and Takeda.

### References

- Holstein SA, Avet-Loiseau H, Hahn T, Ho CM, Lohr JG, Munshi NCet al.BMT CTN Myeloma Intergroup Workshop on Minimal Residual Disease and Immune Profiling: Summary and Recommendations from the Organizing Committee. Biol Blood Marrow Transplant2018; 24(4): 641–648. e-pub ahead of print 2017/12/16; doi: 10.1016/j.bbmt.2017.12.774 [PubMed: 29242112]
- Holstein SA, Ye JC, Howard A, Bhutani M, Gormley N, Hahn Tet al.Summary of the Second Annual BMT CTN Myeloma Intergroup Workshop on Minimal Residual Disease and Immune Profiling. Biol Blood Marrow Transplant2019; 25(3): e89–e97. e-pub ahead of print 2018/11/09; doi: 10.1016/j.bbmt.2018.11.001 [PubMed: 30408566]
- Holstein SA, Al-Kadhimi Z, Costa LJ, Hahn T, Hari P, Hillengass Jet al.Summary of the Third Annual Blood and Marrow Transplant Clinical Trials Network Myeloma Intergroup Workshop on Minimal Residual Disease and Immune Profiling. Biol Blood Marrow Transplant2020; 26(1): e7– e15. e-pub ahead of print 2019/09/19; doi: 10.1016/j.bbmt.2019.09.015 [PubMed: 31526843]

- 4. Holstein SA, Howard A, Avigan D, Bhutani M, Cohen AD, Costa LJet al.Summary of the 2019 Blood and Marrow Transplant Clinical Trials Network Myeloma Intergroup Workshop on Minimal Residual Disease and Immune Profiling. Biol Blood Marrow Transplant2020; 26(10): e247–e255. e-pub ahead of print 2020/06/27; doi: 10.1016/j.bbmt.2020.06.011 [PubMed: 32589921]
- Kumar S, Paiva B, Anderson KC, Durie B, Landgren O, Moreau Pet al.International Myeloma Working Group consensus criteria for response and minimal residual disease assessment in multiple myeloma. Lancet Oncol2016; 17(8): e328–346. doi: 10.1016/S1470-2045(16)30206-6 [PubMed: 27511158]
- Stetler-Stevenson M, Paiva B, Stoolman L, Lin P, Jorgensen JL, Orfao Aet al.Consensus guidelines for myeloma minimal residual disease sample staining and data acquisition. Cytometry B Clin Cytom2016; 90(1): 26–30. doi: 10.1002/cyto.b.21249 [PubMed: 25907102]
- Munshi NC, Avet-Loiseau H, Rawstron AC, Owen RG, Child JA, Thakurta Aet al.Association of Minimal Residual Disease With Superior Survival Outcomes in Patients With Multiple Myeloma: A Meta-analysis. JAMA Oncol2017; 3(1): 28–35. doi: 10.1001/jamaoncol.2016.3160 [PubMed: 27632282]
- Landgren O, Devlin S, Boulad M, Mailankody S. Role of MRD status in relation to clinical outcomes in newly diagnosed multiple myeloma patients: a meta-analysis. Bone Marrow Transplant2016; 51(12): 1565–1568. e-pub ahead of print 2016/09/07; doi: 10.1038/bmt.2016.222 [PubMed: 27595280]
- Munshi NC, Avet-Loiseau H, Anderson KC, Neri P, Paiva B, Samur Met al.A large meta-analysis establishes the role of MRD negativity in long-term survival outcomes in patients with multiple myeloma. Blood Adv2020; 4(23): 5988–5999. e-pub ahead of print 2020/12/08; doi: 10.1182/ bloodadvances.2020002827 [PubMed: 33284948]
- Avet-Loiseau H, Ludwig H, Landgren O, Paiva B, Morris C, Yang Het al.Minimal Residual Disease Status as a Surrogate Endpoint for Progression-free Survival in Newly Diagnosed Multiple Myeloma Studies: A Meta-analysis. Clin Lymphoma Myeloma Leuk2020; 20(1): e30–e37. e-pub ahead of print 2019/11/30; doi: 10.1016/j.clml.2019.09.622 [PubMed: 31780415]
- Paiva B, Cedena MT, Puig N, Arana P, Vidriales MB, Cordon Let al.Minimal residual disease monitoring and immune profiling in multiple myeloma in elderly patients. Blood2016; 127(25): 3165–3174. e-pub ahead of print 2016/04/28; doi: 10.1182/blood-2016-03-705319 [PubMed: 27118453]
- Ho CM, McCarthy PL, Wallace PK, Zhang Y, Fora A, Mellors Pet al.Immune signatures associated with improved progression-free and overall survival for myeloma patients treated with AHSCT. Blood Advances2017; 1(15): 1056–1066. doi: 10.1182/bloodadvances.2017005447 [PubMed: 29296748]
- Bhutani M, Foureau D, Zhang Q, Robinson M, Wynn AS, Steuerwald NMet al.Peripheral Immunotype Correlates with Minimal Residual Disease Status and Is Modulated by Immunomodulatory Drugs in Multiple Myeloma. Biol Blood Marrow Transplant2019; 25(3): 459– 465. e-pub ahead of print 2018/11/28; doi: 10.1016/j.bbmt.2018.11.015 [PubMed: 30481597]
- 14. Lucas F, Pennell M, Huang Y, Benson DM, Efebera YA, Chaudhry Met al. T Cell Transcriptional Profiling and Immunophenotyping Uncover LAG3 as a Potential Significant Target of Immune Modulation in Multiple Myeloma. Biol Blood Marrow Transplant2020; 26(1): 7–15. e-pub ahead of print 2019/08/25; doi: 10.1016/j.bbmt.2019.08.009 [PubMed: 31445183]
- 15. Papadimitriou K, Tsakirakis N, Malandrakis P, Vitsos P, Metousis A, Orologas-Stavrou Net al.Deep Phenotyping Reveals Distinct Immune Signatures Correlating with Prognostication, Treatment Responses, and MRD Status in Multiple Myeloma. Cancers (Basel)2020; 12(11). e-pub ahead of print 2020/11/08; doi: 10.3390/cancers12113245
- Parmar H, Gertz M, Anderson EI, Kumar S, Kourelis TV. Microenvironment immune reconstitution patterns correlate with outcomes after autologous transplant in multiple myeloma. Blood Advances2021; 5(7): 1797–1804. doi: 10.1182/bloodadvances.2020003857 [PubMed: 33787859]
- Munshi NC, Anderson LD Jr., Shah N, Madduri D, Berdeja J, Lonial Set al.Idecabtagene Vicleucel in Relapsed and Refractory Multiple Myeloma. N Engl J Med2021; 384(8): 705–716. e-pub ahead of print 2021/02/25; doi: 10.1056/NEJMoa2024850 [PubMed: 33626253]

- Merz AMA, Merz M, Zhang Y, Stecklein K, Pleskow J, Chen GLet al.Serological Response to Vaccination after Autologous Transplantation for Multiple Myeloma Is Associated with Improved Progression-Free and Overall Survival. Transplant Cell Ther2021; 27(3): 245.e241–245.e248. doi: 10.1016/j.jtct.2020.11.009
- Freidlin B, Korn EL. Biomarker enrichment strategies: matching trial design to biomarker credentials. Nat Rev Clin Oncol2014; 11(2): 81–90. e-pub ahead of print 2013/11/28; doi: 10.1038/nrclinonc.2013.218 [PubMed: 24281059]
- Dobbin KK, Cesano A, Alvarez J, Hawtin R, Janetzki S, Kirsch Iet al. Validation of biomarkers to predict response to immunotherapy in cancer: Volume II - clinical validation and regulatory considerations. J Immunother Cancer2016; 4: 77. e-pub ahead of print 2016/11/29; doi: 10.1186/ s40425-016-0179-0 [PubMed: 27891226]
- Gay F, Musto P, Rota Scalabrini D, Galli M, Belotti A, Zamagni Eet al.Survival Analysis of Newly Diagnosed Transplant-Eligible Multiple Myeloma Patients in the Randomized Forte Trial. Blood2020; 136(Supplement 1): 35–37. doi: 10.1182/blood-2020-136907
- 22. Oliva S, Genuardi E, Petrucci MT, D'Agostino M, Auclair D, Spadano Aet al.Impact of Minimal Residual Disease (MRD) By Multiparameter Flow Cytometry (MFC) and Next-Generation Sequencing (NGS) on Outcome: Results of Newly Diagnosed Transplant-Eligible Multiple Myeloma (MM) Patients Enrolled in the Forte Trial. Blood2020; 136(Supplement 1): 44–45. doi: 10.1182/blood-2020-137351
- 23. Bertamini L, Grasso M, D'Agostino M, Pascarella A, Tosi P, Monaco Fet al.Poor Prognosis of Multiple Myeloma Predicted By High Levels of Circulating Plasma Cells Is Independent from Other High-Risk Features but Is Modulated By the Achievement of Minimal Residual Disease Negativity. Blood2020; 136(Supplement 1): 12–13. doi: 10.1182/blood-2020-137075
- 24. Gay F, Cerrato C, Rota Scalabrini D, Galli M, Belotti A, Zamagni Eet al.Carfilzomib-Lenalidomide-Dexamethasone (KRd) Induction-Autologous Transplant (ASCT)-Krd Consolidation Vs KRd 12 Cycles Vs Carfilzomib-Cyclophosphamide-Dexamethasone (KCd) Induction-ASCT-KCd Consolidation: Analysis of the Randomized Forte Trial in Newly Diagnosed Multiple Myeloma (NDMM). Blood2018; 132(Suppl 1): 121–121. doi: 10.1182/ blood-2018-99-112093
- 25. Oben B, Froyen G, Maclachlan KH, Zheng-Lin B, Yellapantula V, Abascal Fet al.Whole-Genome Sequencing Reveals Evidence of Two Biologically and Clinically Distinct Entities: Progressive Versus Stable Myeloma Precursor Disease. Blood2020; 136(Supplement 1): 47–48. doi: 10.1182/ blood-2020-136403
- 26. Bolli N, Maura F, Minvielle S, Gloznik D, Szalat R, Fullam Aet al.Genomic patterns of progression in smoldering multiple myeloma. Nat Commun2018; 9(1): 3363. e-pub ahead of print 2018/08/24; doi: 10.1038/s41467-018-05058-y [PubMed: 30135448]
- Mateos MV, Kumar S, Dimopoulos MA, González-Calle V, Kastritis E, Hajek Ret al.International Myeloma Working Group risk stratification model for smoldering multiple myeloma (SMM). Blood Cancer J2020; 10(10): 102. e-pub ahead of print 2020/10/18; doi: 10.1038/s41408-02000366-3 [PubMed: 33067414]
- Bustoros M, Sklavenitis-Pistofidis R, Park J, Redd R, Zhitomirsky B, Dunford AJet al.Genomic Profiling of Smoldering Multiple Myeloma Identifies Patients at a High Risk of Disease Progression. J Clin Oncol2020; 38(21): 2380–2389. e-pub ahead of print 2020/05/23; doi: 10.1200/JCO.20.00437 [PubMed: 32442065]
- Da Via MC, Ziccheddu B, Maeda A, Bagnoli F, Perrone G, Bolli N. A Journey Through Myeloma Evolution: From the Normal Plasma Cell to Disease Complexity. Hemasphere2020; 4(6): e502.
   e-pub ahead of print 2020/12/08; doi: 10.1097/HS9.000000000000502 [PubMed: 33283171]
- Zavidij O, Haradhvala NJ, Mouhieddine TH, Sklavenitis-Pistofidis R, Cai S, Reidy Met al.Singlecell RNA sequencing reveals compromised immune microenvironment in precursor stages of multiple myeloma. Nat Cancer2020; 1(5): 493–506. e-pub ahead of print 2021/01/08; doi: 10.1038/s43018-020-0053-3 [PubMed: 33409501]
- Bernardini G, Sciume G, Santoni A. Differential chemotactic receptor requirements for NK cell subset trafficking into bone marrow. Front Immunol2013; 4: 12. e-pub ahead of print 2013/02/07; doi: 10.3389/fimmu.2013.00012 [PubMed: 23386850]

- Ponzetta A, Benigni G, Antonangeli F, Sciume G, Sanseviero E, Zingoni Aet al.Multiple Myeloma Impairs Bone Marrow Localization of Effector Natural Killer Cells by Altering the Chemokine Microenvironment. Cancer Res2015; 75(22): 4766–4777. e-pub ahead of print 2015/10/07; doi: 10.1158/0008-5472.CAN-15-1320 [PubMed: 26438594]
- Kawano Y, Zavidij O, Park J, Moschetta M, Kokubun K, Mouhieddine THet al.Blocking IFNAR1 inhibits multiple myeloma-driven Treg expansion and immunosuppression. J Clin Invest2018; 128(6): 2487–2499. e-pub ahead of print 2018/03/21; doi: 10.1172/jci88169 [PubMed: 29558366]
- Ledergor G, Weiner A, Zada M, Wang SY, Cohen YC, Gatt MEet al.Single cell dissection of plasma cell heterogeneity in symptomatic and asymptomatic myeloma. Nat Med2018; 24(12): 1867–1876. e-pub ahead of print 2018/12/14; doi: 10.1038/s41591-018-0269-2 [PubMed: 30523328]
- Merz M, Merz AMA, Wang J, Wei L, Belal A, Alberico Ret al.Spatiotemporal Assessment of Immunogenomic Heterogeneity in Multiple Myeloma. Blood2020; 136(Supplement 1): 14–15. doi: 10.1182/blood-2020-142622
- 36. Calcinotto A, Brevi A, Chesi M, Ferrarese R, Garcia Perez L, Grioni Met al.Microbiotadriven interleukin-17-producing cells and eosinophils synergize to accelerate multiple myeloma progression. Nat Commun2018; 9(1): 4832. e-pub ahead of print 2018/12/05; doi: 10.1038/ s41467-018-07305-8 [PubMed: 30510245]
- Ivanov II, Atarashi K, Manel N, Brodie EL, Shima T, Karaoz Uet al.Induction of intestinal Th17 cells by segmented filamentous bacteria. Cell2009; 139(3): 485–498. e-pub ahead of print 2009/10/20; doi: 10.1016/j.cell.2009.09.033 [PubMed: 19836068]
- 38. Pianko MJ, Devlin SM, Littmann ER, Chansakul A, Mastey D, Salcedo Met al.Minimal residual disease negativity in multiple myeloma is associated with intestinal microbiota composition. Blood Adv2019; 3(13): 2040–2044. e-pub ahead of print 2019/07/11; doi: 10.1182/ bloodadvances.2019032276 [PubMed: 31289031]
- Lust JA, Lacy MQ, Zeldenrust SR, Witzig TE, Moon-Tasson LL, Dinarello CAet al.Reduction in C-reactive protein indicates successful targeting of the IL-1/IL-6 axis resulting in improved survival in early stage multiple myeloma. Am J Hematol2016; 91(6): 571–574. e-pub ahead of print 2016/03/08; doi: 10.1002/ajh.24352 [PubMed: 26945843]
- Paiva B, Puig N, Cedena MT, Rosiñol L, Cordón L, Vidriales MBet al.Measurable Residual Disease by Next-Generation Flow Cytometry in Multiple Myeloma. J Clin Oncol2020; 38(8): 784–792. e-pub ahead of print 2019/11/27; doi: 10.1200/jco.19.01231 [PubMed: 31770060]
- Botta C, Pérez Ruiz C, Goicoechea I, Puig N, Cedena MT, Cordon Let al.Single-Cell Characterization of the Multiple Myeloma (MM) Immune Microenvironment Identifies CD27-Negative T Cells As Potential Source of Tumor-Reactive Lymphocytes. Blood2019; 134(Supplement\_1): 506–506. doi: 10.1182/blood-2019-123974
- 42. Puig N, Mateos M-V, Contreras T, Paiva B, Cedena MT, Pérez JJet al.Qip-Mass Spectrometry in High Risk Smoldering Multiple Myeloma Patients Included in the GEM-CESAR Trial: Comparison with Conventional and Minimal Residual Disease IMWG Response Assessment. Blood2019; 134(Supplement\_1): 581–581. doi: 10.1182/blood-2019-127717
- 43. Eveillard M, Rustad E, Roshal M, Zhang Y, Ciardiello A, Korde Net al.Comparison of MALDI-TOF mass spectrometry analysis of peripheral blood and bone marrow-based flow cytometry for tracking measurable residual disease in patients with multiple myeloma. Br J Haematol2020; 189(5): 904–907. e-pub ahead of print 2020/02/07; doi: 10.1111/bjh.16443 [PubMed: 32026474]
- 44. Shah V, Sherborne AL, Walker BA, Johnson DC, Boyle EM, Ellis Set al.Prediction of outcome in newly diagnosed myeloma: a meta-analysis of the molecular profiles of 1905 trial patients. Leukemia2018; 32(1): 102–110. e-pub ahead of print 2017/06/07; doi: 10.1038/leu.2017.179 [PubMed: 28584253]
- Fan J, Slowikowski K, Zhang F. Single-cell transcriptomics in cancer: computational challenges and opportunities. Exp Mol Med2020; 52(9): 1452–1465. e-pub ahead of print 2020/09/16; doi: 10.1038/s12276-020-0422-0 [PubMed: 32929226]
- 46. Satpathy AT, Granja JM, Yost KE, Qi Y, Meschi F, McDermott GPet al.Massively parallel single-cell chromatin landscapes of human immune cell development and intratumoral T cell exhaustion. Nat Biotechnol2019; 37(8): 925–936. e-pub ahead of print 2019/08/04; doi: 10.1038/ s41587-019-0206-z [PubMed: 31375813]

- 47. Galletti G, De Simone G, Mazza EMC, Puccio S, Mezzanotte C, Bi TMet al.Two subsets of stem-like CD8(+) memory T cell progenitors with distinct fate commitments in humans. Nat Immunol2020; 21(12): 1552–1562. e-pub ahead of print 2020/10/14; doi: 10.1038/ s41590-020-0791-5 [PubMed: 33046887]
- 48. Crinier A, Milpied P, Escaliere B, Piperoglou C, Galluso J, Balsamo Aet al.High-Dimensional Single-Cell Analysis Identifies Organ-Specific Signatures and Conserved NK Cell Subsets in Humans and Mice. Immunity2018; 49(5): 971–986 e975. e-pub ahead of print 2018/11/11; doi: 10.1016/j.immuni.2018.09.009 [PubMed: 30413361]
- Neri P, Maity R, Tagoug I, McCulloch S, Duggan P, Jimenez-Zepeda Vet al.Immunome Single Cell Profiling Reveals T Cell Exhaustion with Upregulation of Checkpoint Inhibitors LAG3 and Tigit on Marrow Infiltrating T Lymphocytes in Daratumumab and IMiDs Resistant Patients. Blood2018; 132(Suppl 1): 242–242. doi: 10.1182/blood-2018-99-117531
- 50. Leblay N, Maity R, Barakat E, McCulloch S, Duggan P, Jimenez-Zepeda Vet al.Cite-Seq Profiling of T Cells in Multiple Myeloma Patients Undergoing BCMA Targeting CAR-T or Bites Immunotherapy. Blood2020; 136(Supplement 1): 11–12. doi: 10.1182/blood-2020-137650 [PubMed: 32276273]
- Bailur JK, McCachren SS, Doxie DB, Shrestha M, Pendleton K, Nooka AKet al.Early alterations in stem-like/resident T cells, innate and myeloid cells in the bone marrow in preneoplastic gammopathy. JCI Insight2019; 5. e-pub ahead of print 2019/04/24; doi: 10.1172/jci.insight.127807
- Fionda C, Stabile H, Molfetta R, Soriani A, Bernardini G, Zingoni Aet al.Translating the antimyeloma activity of Natural Killer cells into clinical application. Cancer Treat Rev2018; 70: 255– 264. e-pub ahead of print 2018/10/17; doi: 10.1016/j.ctrv.2018.10.005 [PubMed: 30326421]
- Shah N, Mehta R, Li L, Mccarty J, Kaur I, Orlowski RZet al.Phase II study of ex vivo expanded cord blood natural killer cells for multiple myeloma. Journal of Clinical Oncology2018; 36(15\_suppl): 8006–8006. doi: 10.1200/JCO.2018.36.15\_suppl.8006
- 54. Sarkar S, O'Ceallaigh C, O'Dwyer M. Genetically modified natural killer cells: an "off-the-shelf" immunotherapy for cancer. Cell Gene Therapy Insights2019; 5(5).
- 55. Pomeroy EJ, Hunzeker JT, Kluesner MG, Lahr WS, Smeester BA, Crosby MRet al.A Genetically Engineered Primary Human Natural Killer Cell Platform for Cancer Immunotherapy. Mol Ther2020; 28(1): 52–63. e-pub ahead of print 2019/11/11; doi: 10.1016/j.ymthe.2019.10.009 [PubMed: 31704085]
- 56. Naeimi Kararoudi M, Nagai Y, Elmas E, de Souza Fernandes Pereira M, Ali SA, Imus PHet al.CD38 deletion of human primary NK cells eliminates daratumumab-induced fratricide and boosts their effector activity. Blood2020; 136(21): 2416–2427. e-pub ahead of print 2020/07/01; doi: 10.1182/blood.2020006200 [PubMed: 32603414]
- 57. Liu E, Marin D, Banerjee P, Macapinlac HA, Thompson P, Basar Ret al.Use of CAR-Transduced Natural Killer Cells in CD19-Positive Lymphoid Tumors. N Engl J Med2020; 382(6): 545–553. e-pub ahead of print 2020/02/06; doi: 10.1056/NEJMoa1910607 [PubMed: 32023374]
- Leivas A, Risueno RM, Guzman A, Sanchez-Vega L, Perez M, Megias Det al.Natural killer cells efficiently target multiple myeloma clonogenic tumor cells. Cancer Immunol Immunother2021. e-pub ahead of print 2021/03/12; doi: 10.1007/s00262-021-02901-y
- Schmidts A, Ormhoj M, Choi BD, Taylor AO, Bouffard AA, Scarfo Iet al.Rational design of a trimeric APRIL-based CAR-binding domain enables efficient targeting of multiple myeloma. Blood Adv2019; 3(21): 3248–3260. e-pub ahead of print 2019/11/08; doi: 10.1182/ bloodadvances.2019000703 [PubMed: 31698455]
- Anagnostou T, Riaz IB, Hashmi SK, Murad MH, Kenderian SS. Anti-CD19 chimeric antigen receptor T-cell therapy in acute lymphocytic leukaemia: a systematic review and meta-analysis. Lancet Haematol2020; 7(11): e816–e826. e-pub ahead of print 2020/10/23; doi: 10.1016/ s2352-3026(20)30277-5 [PubMed: 33091355]
- Lemal R, Tournilhac O. State-of-the-art for CAR T-cell therapy for chronic lymphocytic leukemia in 2019. J Immunother Cancer2019; 7(1): 202. e-pub ahead of print 2019/08/03; doi: 10.1186/ s40425-019-0686-x [PubMed: 31370892]
- 62. Fraietta JA, Lacey SF, Orlando EJ, Pruteanu-Malinici I, Gohil M, Lundh Set al.Determinants of response and resistance to CD19 chimeric antigen receptor (CAR) T cell therapy of chronic

lymphocytic leukemia. Nat Med2018; 24(5): 563–571. e-pub ahead of print 2018/05/02; doi: 10.1038/s41591-018-0010-1 [PubMed: 29713085]

- 63. Garfall AL, Dancy EK, Cohen AD, Hwang WT, Fraietta JA, Davis MMet al.T-cell phenotypes associated with effective CAR T-cell therapy in postinduction vs relapsed multiple myeloma. Blood Adv2019; 3(19): 2812–2815. e-pub ahead of print 2019/10/03; doi: 10.1182/ bloodadvances.2019000600 [PubMed: 31575532]
- 64. A PHASE 1 STUDY OF CT103A, A FULLY HUMAN BCMA TARGETING CAR T CELL, IN SUBJECTS WITH RELAPSED/REFRACTORY MULTIPLE MYELOMA. EHA2020, 2020.
- 65. Murthy HS, Gharaibeh RZ, Al-Mansour Z, Kozlov A, Trikha G, Newsome RCet al.Baseline Gut Microbiota Composition Is Associated with Major Infections Early after Hematopoietic Cell Transplantation. Biol Blood Marrow Transplant2020; 26(11): 2001–2010. e-pub ahead of print 2020/07/28; doi: 10.1016/j.bbmt.2020.07.023 [PubMed: 32717434]
- 66. Tanaka JS, Young RR, Heston SM, Jenkins K, Spees LP, Sung ADet al.Anaerobic Antibiotics and the Risk of Graft-versus-Host Disease after Allogeneic Hematopoietic Stem Cell Transplantation. Biol Blood Marrow Transplant2020; 26(11): 2053–2060. e-pub ahead of print 2020/07/20; doi: 10.1016/j.bbmt.2020.07.011 [PubMed: 32682948]
- 67. Konuma T, Kohara C, Watanabe E, Takahashi S, Ozawa G, Inomata Ket al.Impact of Intestinal Microbiota on Reconstitution of Circulating Monocyte, Dendritic Cell, and Natural Killer Cell Subsets in Adults Undergoing Single-Unit Cord Blood Transplantation. Biol Blood Marrow Transplant2020; 26(11): e292–e297. e-pub ahead of print 2020/08/18; doi: 10.1016/ j.bbmt.2020.08.009 [PubMed: 32798658]
- Abid MB, Shah NN, Maatman TC, Hari PN. Gut microbiome and CAR-T therapy. Exp Hematol Oncol2019; 8: 31. e-pub ahead of print 2019/12/13; doi: 10.1186/s40164-019-0155-8 [PubMed: 31827982]
- Smith M, Littmann ER, Slingerland JB, Clurman A, Slingerland AE, Taur Yet al.Intestinal Microbiota Composition Prior to CAR T Cell Infusion Correlates with Efficacy and Toxicity. Blood2018; 132(Supplement 1): 3492–3492. doi: 10.1182/blood-2018-99-118628
- Chakraborty R, Hill BT, Majeed A, Majhail NS. Late Effects after Chimeric Antigen Receptor T Cell Therapy for Lymphoid Malignancies. Transplant Cell Ther2021; 27(3): 222–229. doi: 10.1016/j.jtct.2020.10.002 [PubMed: 33928266]
- Sanchez CE, Dowlati EP, Geiger AE, Chaudhry K, Tovar MA, Bollard CMet al.NK Cell Adoptive Immunotherapy of Cancer: Evaluating Recognition Strategies and Overcoming Limitations. Transplant Cell Ther2021; 27(1): 21–35. doi: 10.1016/j.bbmt.2020.09.030 [PubMed: 33007496]

### Highlights

- Comprehensive summary of the 2020 BMT CTN Myeloma Intergroup
  MRD/IP workshop
- Real world use of MRD testing is limited by lack of evidence-based guidelines
- Challenges exist in establishing MRD as a surrogate endpoint
- Molecular and IP studies reveal insight into disease biology and treatment response
- Cellular therapy is evolving beyond autologous BCMA CAR T-cells